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Abstract 

In order to make proton exchange membrane fuel cell (PEMFC) more economically viable, 

researchers have been developing non-noble metal catalyst to replace platinum catalyst, 

redesigning PEMFC to increase its efficiency and studying various performance loss causing 

issues like water accumulation, poisoning of catalyst, membrane electrode assembly (MEA) 

degradation. In the present work, the focus has been on diagnosing and mitigating performance 

loss causing issues like water accumulation in cathode channel and CO poisoning of anode 

electro-catalyst. Mathematical modelling and simulation has also been carried out to better 

understand the mechanism behind CO poisoning of anode electro-catalyst and its regeneration 

process. 

A single channel PEMFC with bends at the entrance and exit has been used to investigate 

effects on the performance of PEMFC due water accumulation in cathode channel. For this 

purpose, MEAs with reversible hydrogen electrodes (RHE) are placed in a series of sub cells 

and current-voltage (i-V) characteristics and electrochemical impedance spectroscopy (EIS) 

data are collected for anode and cathode separately. By comparing i-V characteristics and EIS 

data between different sub cells, a dip in i-V curve is found to follow cathode polarization loss 

from sub cell 1 to 5. The increase in charge transfer resistance at the cathode and dip in i-V 

curve attributed to product water accumulation in subsequent sub cells. However, no dip in i-

V characteristics and increase in cathode charge transfer resistance is observed when single sub 

cells operated at different locations, i.e., close to entry, middle and exit of the channel. Further, 

in sub cell 5, accumulation of product water at the cathode and in the exit bend leads to drastic 

reduction in the limiting current density due to high concentration over-potential.  

In case of CO poisoning study of anode electro-catalyst, an experimental procedure is 

developed based on break-in procedure and dilute KMnO4 solution is used to regenerate the 
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CO poisoned Pt black anode electro-catalyst of a PEMFC. Also to understand the effect of 

operating temperature on extent of CO poisoning and regeneration of CO poisoned Pt black 

anode electro-catalyst, PEMFC is operated at 30 ⁰C, 50 ⁰C and 70 ⁰C. At 50 ⁰C and 70 ⁰C, 

break-in procedure contributes towards recovering the initial performance i.e. 74.6 % and 78.8 

% of the initial current density at 0.2 V respectively and rest is regenerated by use of dilute 

KMnO4 solution. Later, in order to restrict the usage of dilute KMnO4 solution, the 

experimental procedure is further modified by increasing the number of activation cycle. 

In order to better understand the actual mechanism being followed during CO poisoning and 

regeneration of CO poisoned electro-catalyst, a single cell PEMFC mathematical model 

coupled with surface coverage equations are simulated. Hydrogen oxidation reaction (HOR) 

are assumed to follow Heyrovsky-Volmer mechanism “before CO poisoning” and “after 

activation cycle” and Tafel-Volmer mechanisms “after CO poisoning” stage of operations. 

During CO poisoning, surface coverage variation for CO and H2 clearly shows that adsorption 

of CO hinders adsorption of H2 and eventually decreases the cell performance. The above 

mentioned diagnostics and regeneration procedures followed for water accumulation and CO 

poisoning can be used extend to large scale PEMFC stacks, which may facilitate PEMFC 

commercialisation.  
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सार 
 

प्रोटॉन एक्सचेंज मेमे्ब्रन फू्यल सेल (PEMFC) को अधिक आधथिक रूप से व्यवहार्ि बनाने के धलए, 

शोिकर्ािओ ंने पै्लधटनम उते्प्ररक को बदलने के धलए नॉन-नोबल मेटल उते्प्ररक धवकधसर् धकर्ा है, इसकी 

दक्षर्ा बढाने के धलए PEMFC को धिर से धिजाइन धकर्ा गर्ा है और जल संचर्, उते्प्ररक की धवषाक्तर्ा 

के कारण धवधिन्न प्रदशिन हाधन का अध्यर्न धकर्ा है। , धिल्ली इलेक्ट्र ोि असेंबली (MEA) की धगरावट। 

वर्िमान कार्ि में, प्रदशिन हाधन को कम करने और कैथोि चैनल में जल संचर् और एनोि इलेक्ट्र ो-उते्प्ररक 

के सीओ धवषाक्तर्ा जैसे मुद्ो ंका कारण बनर्ा है। एनोि इलेक्ट्र ो-उते्प्ररक और इसके पुनजिनन प्रधिर्ा 

के सीओ धवषाक्तर्ा के पीछे के रं्त्र को बेहर्र ढंग से समिने के धलए गधणर्ीर् मॉिधलंग और धसमुलेशन 

िी धकर्ा गर्ा है। 

प्रवेश द्वार पर िुकर्ा हुआ एक एकल चैनल PEMFC और कैथोि चैनल में पानी के जमाव के कारण 

PEMFC के प्रदशिन पर पड़ने वाले प्रिावो ंकी जांच करने के धलए उपर्ोग धकर्ा गर्ा है। इस प्रर्ोजन के 

धलए, प्रधर्वर्ती हाइिर ोजन इलेक्ट्र ोि (RHE) वाले MEAs को उप कोधशकाओ ंकी एक श्रंखला में रखा 

जार्ा है और वर्िमान-वोले्टज (i-V) धवशेषर्ाओ ंऔर धवद्रु्र् रासार्धनक प्रधर्बािा से्पक्ट्र ोस्कोपी (EIS) 

िेटा को अलग से एनोि और कैथोि के धलए एकत्र धकर्ा जार्ा है। धवधिन्न उप कोधशकाओ ंके बीच 

आईवी धवशेषर्ाओ ंऔर ईआईएस िेटा की रु्लना करके, आईवी वि में एक िुबकी उप सेल 1 से 5. से 

कैथोि धु्रवीकरण नुकसान का पालन करने के धलए पार्ा जार्ा है। कैथोि में प्रिारी स्थानांर्रण प्रधर्रोि 

में वरद्धि और उत्पाद जल संचर् के धलए धजमे्मदार iV वि में िुबकी। बाद के उप कक्षो ंमें। हालांधक, आई-

वी धवशेषर्ाओ ंऔर कैथोि चाजि टर ांसिर प्रधर्रोि में कोई धगरावट नही ंदेखी जार्ी है, जब धवधिन्न स्थानो ं

पर एकल उप कोधशकाओ ंको संचाधलर् धकर्ा जार्ा है, अथािर्, चैनल के प्रवेश, मध्य और धनकास के 

करीब। इसके अलावा, उप सेल 5 में, कैथोि पर और धनकास मोड़ पर उत्पाद के पानी का संचर् उच्च 

सांद्रर्ा क्षमर्ा के कारण सीधमर् घनत्व में िारी कमी की ओर जार्ा है। 

एनोि इलेक्ट्र ो-उते्प्ररक के सीओ धवषाक्तर्ा अध्यर्न के मामले में, एक प्रार्ोधगक प्रधिर्ा बे्रक-इन प्रधिर्ा 

के आिार पर धवकधसर् की जार्ी है और एक पीईएमएिसी के सीओ जहर पीटी काले एनोि इलेक्ट्र ो-

उते्प्ररक को पुन: उत्पन्न करने के धलए KMnO4 समािान का उपर्ोग धकर्ा जार्ा है। सीओ धवषाक्तर्ा 

की सीमा पर ऑपरेधटंग र्ापमान के प्रिाव को समिने के धलए और सीओ जहर वाले पं। काले एनोि 

इलेक्ट्र ो-उते्प्ररक के पुनजिनन को समिने के धलए, पीईएमएिसी को 30 °C, 50 °C और 70 °C पर 

संचाधलर् धकर्ा जार्ा है। 50 °C और 70 °C पर, बे्रक-इन प्रधिर्ा प्रारंधिक प्रदशिन र्ानी 74.6% और 

78.8% के प्रारंधिक वर्िमान घनत्व को िमशः 0.2 V पर पुनप्रािप्त करने की धदशा में र्ोगदान करर्ी है 

और पर्ला KMnO4 समािान के उपर्ोग से पुनजतीधवर् होर्ी है। बाद में, पर्ला KMnO4 समािान के 

उपर्ोग को प्रधर्बंधिर् करने के धलए, सधिर्ण चि की संख्या में वरद्धि करके प्रर्ोगात्मक प्रधिर्ा को 

और संशोधिर् धकर्ा गर्ा है। 

सीओ धवषाक्तर्ा और सीओ जहर इलेक्ट्र ो-उते्प्ररक के उत्थान के दौरान पीछा धकए जाने वाले वास्तधवक 

रं्त्र को बेहर्र ढंग से समिने के धलए, सर्ह कवरेज समीकरणो ं के साथ धमलकर एक एकल सेल 

पीईएमएिसी गधणर्ीर् मॉिल को धसमु्यलेटेि धकर्ा जार्ा है। हाइिर ोजन ऑक्सीकरण प्रधर्धिर्ा (HOR) 

को माना जार्ा है धक हेओरोव्स्स्की-वोल्मर रं्त्र "सीओ धवषाक्तर्ा से पहले" और "सधिर्ण चि के बाद" 

और टािेल-वोल्मर रं्त्र "सीओ धवषाक्तर्ा के बाद" संचालन के चरण का पालन करें । सीओ धवषाक्तर्ा 

के दौरान, सीओ और एच 2 के धलए सर्ह कवरेज धिन्नर्ा स्पष्ट रूप से धदखार्ी है धक सीओ का सोखना 

एच 2 के सोखना में बािा िालर्ा है और अंर्र्ः सेल के प्रदशिन को कम करर्ा है। पानी के संचर् और 

CO धवषाक्तर्ा के धलए उपरु्िक्त धनदान और पुनजिनन प्रधिर्ाओ ंका उपर्ोग बडे़ पैमाने पर PEMFC 

सै्टक र्क धकर्ा जा सकर्ा है, जो PEMFC व्यावसार्ीकरण की सुधविा प्रदान कर सकर्ा है। 
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